Author's response to reviews

Title: The role of c-Src in the invasion and metastasis of hepatocellular carcinoma cells induced by association of cell surface GRP78 with activated alpha2M

Authors:

song zhao (Zhaosong517@163.com)
hongdan li (lihongdan101@126.com)
qingjun wang (qingjunwang666@163.com)
chang su (changsu80@163.com)
guan wang (Wang591716708@163.com)
huijuan song (shi-shi@tom.com)
liang zhao (liangzhao79@163.com)
luan zhidong (zhluanqwe@hotmail.com)
rongjian su (rongjiansu@hotmail.com)

Version: 2 Date: 13 November 2014

Author's response to reviews:

This manuscript has been initially subjected to molecular cancer. After editotrial evaluation, they think that this manuscript is not suitable for punication in molecular cancer and suggested that we transfer it ti your journal.